Reviewer #1:

Specific Comments To Authors: Manuscript ID: 85877 - World Journal of Clinical Cases Manuscript Title: Identification of Key Genes and Biological Pathways in Lung Adenocarcinoma by Integrated Bioinformatics Analysis

Dear authors, The study is interesting and it was well conducted. In order to contribute to the improvement of the manuscript, some suggestions are proposed according to the Guideline for Basic study: 1. Please note that the manuscript must be prepared using 12 pt Book Antiqua font. Do not use superscript numbers or symbols to identify the authors' affiliation; 2. ORCID number of the authors is required; 3. In the abstract, in RESULTS, the authors state that “The low expression of these genes was associated with poor prognosis in LUAD”. Wouldn't it be the increase of expression? Best regards

Answer:
1. The manuscript has been changed to 12 pt Book Antiqua font. Removed superscript numbers or symbols that identify author affiliation.

2. All authors' ORCID numbers have been added.

3. In the abstract, in RESULTS, it should be “ The high expression of these genes was associated with poor prognosis in LUAD. “ This is our negligence. Thank you for reminding us.

Reviewer #2:

Specific Comments To Authors: The authors review the gene expression of lung adenocarcinoma. The manuscript is written. However, I some concerns to discuss. -Is the review narrative or systematic? -How about gene expression

**Answer:**

1. This is a systematic review.

2. The expression of genes during the transfer process is currently unclear, which is our future research direction.